Mauna Loa Solar Observatory Observer's Log ______ Wed Jun 4 16:16:24 GMT 2014 Year: 14 Doy: 155 Observer: berkey WEATHER COMMENT: berkey: Wed Jun 4 16:16:30 GMT 2014 temp 38f, wind 5mph, sky mostly clear with thin patchy cirrus scatter across the sky end **PSPT PROBLEM COMMENT BY berkey** : Wed Jun 4 16:17:04 GMT 2014 Pspt xterm as a lot of messages related to NSF server no responding and then later being ok. I also get a "No more proces ses." error when I try to run something. I think this means we are out of virtual memory on the system. Logging out of ope n windows seems to have cleared up the issue for now... end Wed Jun 04 16:31:58 GMT 2014 COMP Start Patrol Wed Jun 4 16:42:59 GMT 2014: PSPT Start Patrol Wed Jun 04 16:44:31 GMT 2014 COMP End Patrol Wed Jun 04 16:45:39 GMT 2014 COMP Start Patrol Wed Jun 04 16:57:06 GMT 2014 KCOR Start Synoptic Patrol Wed Jun 04 17:15:54 GMT 2014 KCOR End Patrol GENERAL OBSERVATORY COMMENT BY berkey: Wed Jun 4 17:16:19 GMT 2014 Haze/cirrus increasing end Wed Jun 04 17:32:59 GMT 2014 COMP End Patrol GENERAL OBSERVATORY COMMENT BY berkey: Wed Jun 4 18:03:14 GMT 2014 main UPS just shut down its load side for about 15 seconds. Many systems were effected, network equipment, sgs, ulua, iwi i all shut down. It is not clear why this hiccup occoured. end GENERAL OBSERVATORY COMMENT BY berkey: Wed Jun 4 23:32:03 GMT 2014 Closing domes end Wed Jun 4 23:32:30 GMT 2014: PSPT Abort Patrol GENERAL OBSERVATORY COMMENT BY berkey: Thu Jun 5 01:29:17 GMT 2014 Big cryo dewar is empty, but we had enough Ln2 to fill comp. We are scheduled for Ln2 delievery today so everything seems to be working out. end GENERAL OBSERVATORY COMMENT BY berkey: Thu Jun 5 01:29:58 GMT 2014 Removed the default gateway tcp/ip config for the printer, hopefully this will stop it from printing random junk from the internet. end